



Limnology and Oceanography Letters 9, 2024, 237–246

© 2024 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA. doi: 10.1002/lol2.10372

### LETTER

# Establishing fluvial silicon regimes and their stability across the Northern Hemisphere

Keira Johnson <sup>(D)</sup>, <sup>1</sup> Kathi Jo Jankowski <sup>(D)</sup>, <sup>2</sup> Joanna Carey <sup>(D)</sup>, <sup>3</sup> Nicholas J. Lyon <sup>(D)</sup>, <sup>4</sup> William H. McDowell <sup>(D)</sup>, <sup>5</sup> Arial Shogren <sup>(D)</sup>, <sup>6</sup>
Adam Wymore <sup>(D)</sup>, <sup>7</sup> Lienne Sethna <sup>(D)</sup>, <sup>8</sup> Wilfred M. Wollheim <sup>(D)</sup>, <sup>7</sup> Amanda E. Poste <sup>(D)</sup>, <sup>9,10</sup> Pirkko Kortelainen <sup>(D)</sup>, <sup>11</sup> Ruth Heindel <sup>(D)</sup>, <sup>12</sup>
Hjalmar Laudon <sup>(D)</sup>, <sup>13</sup> Antti Räike <sup>(D)</sup>, <sup>11</sup> Jeremy B. Jones <sup>(D)</sup>, <sup>14,15</sup> Diane McKnight, <sup>16</sup> Paul Julian <sup>(D)</sup>, <sup>17</sup> Sidney Bush <sup>(D)</sup>, <sup>1</sup> Pamela L. Sullivan <sup>(D)</sup>, <sup>1\*</sup>
<sup>1</sup> College of Earth, Ocean, and Atmospheric Sciences, Oregon State University, Corvallis, Oregon, USA; <sup>2</sup>U.S. Geological Survey, Upper Midwest Environmental Sciences Center, La Crosse, Wisconsin, USA; <sup>3</sup>Math, Analytics, Science and Technology Division, Babson College, Wellesley, Massachusetts, USA; <sup>4</sup>National Center for Ecological Analysis and Synthesis, University of California, Santa Barbara, California, USA; <sup>5</sup>Department of Natural Resources and the Environment, University of New Hampshire, Durham, New Hampshire, USA; <sup>6</sup>Department of Biological Sciences, The University of Alabama, Tuscaloosa, Alabama, USA; <sup>7</sup>Department of Natural Resources and the Environment, University of New Hampshire, USA; <sup>8</sup>St. Croix Watershed Research Station, Marine on St. Croix, Minnesota, USA; <sup>9</sup>Department of Arctic Ecology, Norwegian Institute for Nature Research, Tromsø, Norway; <sup>10</sup>Norwegian Institute for Water Research, Oslo, Norway; <sup>11</sup>Finnish Environment Institute, Helsinki, Finland; <sup>12</sup>Environmental Studies Program, Kenyon College, Gambier, Ohio, USA; <sup>13</sup>Forest Ecology and Management, Swedish University of Agricultural Sciences, Umeå, Sweden; <sup>14</sup>Department of Biology and Wildlife, University of Alaska Fairbanks, Alaska, USA; <sup>15</sup>Institute of Arctic Biology, Fairbanks, Alaska, USA; <sup>16</sup>Department of Civil, Environmental, and Architectural Engineering, University of Colorado Boulder, Boulder, Colorado, USA; <sup>17</sup>The Everglades Foundation, Palmetto Bay, Florida, USA

#### Scientific Significance Statement

Compared to other nutrients, silicon (Si) is understudied despite its importance for primary productivity, water quality, and global carbon sequestration. Understanding Si regimes—the seasonal patterns of Si concentrations—can help identify processes driving fluvial Si concentrations. Regimes describe the annual timing and magnitude of concentrations to identify hydrobiogeochemical, climate, and land use controls on Si concentrations. This article identifies the main Si concentration regimes from over 200 stream sites across the Northern Hemisphere. The results of our analysis indicate more variation in Si seasonality and stability than previously recognized both within and across ecosystems, highlighting the need to understand patterns and drivers related to intraannual and interannual variability in Si cycling.

Data Availability Statement: Data and metadata are available at https://doi.org/10.5066/P9CXDEE8 (Johnson et al. 2023).

Additional Supporting Information may be found in the online version of this article.

<sup>\*</sup>Correspondence: pamela.sullivan@oregonstate.edu

Associate editor: Emily H. Stanley

Author Contribution Statement: K.J., K.J.J., J.C., P.S., and A.S. designed the research. K.J., K.J.J., J.C., N.J.L., W.H.M.D., A.S., A.W., L.S., W.M.W., A.E.P., P.K., R.H., H.L., A.R., J.B.J., D.M.K., P.J., and P.S. provided data to the dataset. K.J., K.J.J., P.J., J.C., and L.S. harmonized and QA/QCed the dataset. K.J., K.J.J., and N.L. performed the analysis. P.S., K.J.J., W.H.M.D., A.S., and A.W. provided guidance and methodological ideas for analysis. K.J.J., J.C., W.H.M.D., A.S., A.W., L.S., W.H.M.D., A.S., A.W., L.S., W.H.M.D., A.S., and A.W. provided guidance and methodological ideas for analysis. K.J.J., J.C., W.H.M.D., A.S., A.W., L.S., W.M.W., A.E.P., P.K., R.H., H.L., A.R., J.B.J., D.M.K., P.J., and P.S. helped interpret results. K.J., K.J.J., and P.S. wrote the manuscript. J.C., N.J.L., W.H.M.D., A.S., A.W., L.S., W.M.W., A.E.P., P.K., R.H., H.L., A.R., J.B.J., D.M.K., P.J., and S.B. reviewed and edited the manuscript.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

#### Abstract

23782242, 2024.3, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/ol2.10372 by Swedish University Of Agricultural Sciences, Wiley Online Library on [03/07/2024]. See the Terms and Conditions

· (https:

onlinelibrary.wiley.com/

Wiley Online Library

for rules of use; OA articles

are governed by the applicable Creative Commons License

Fluvial silicon (Si) plays a critical role in controlling primary production, water quality, and carbon sequestration through supporting freshwater and marine diatom communities. Geological, biogeochemical, and hydrological processes, as well as climate and land use, dictate the amount of Si exported by streams. Understanding Si regimes—the seasonal patterns of Si concentrations—can help identify processes driving Si export. We analyzed Si concentrations from over 200 stream sites across the Northern Hemisphere to establish distinct Si regimes and evaluated how often sites moved among regimes over their period of record. We observed five distinct regimes across diverse stream sites, with nearly 60% of sites exhibiting multiple regime types over time. Our results indicate greater spatial and interannual variability in Si seasonality than previously recognized and highlight the need to characterize the watershed and climate variables that affect Si cycling across diverse ecosystems.

Fluvial silicon (Si) plays a critical role in controlling primary productivity, ecosystem health, and carbon sequestration in freshwater, estuarine, and marine ecosystems (Conley et al. 1993; Buesseler 1998; Gregg and Rousseaux 2014). Si export supports diatom communities, which are responsible for  $\sim 20\%$  of photosynthetically fixed CO<sub>2</sub> each year (Malviva et al. 2016). However, warming and land use change are altering how watersheds cycle Si. Warming increases the silicate weathering rate, which in turn increases Si exports (Deng et al. 2022). Additionally, urbanization may increase Si transport to streams (Carey and Fulweiler 2012; Maguire and Fulweiler 2016) whereas agricultural practices have been linked to reductions in Si export (Struyf et al. 2010). Shifts in Si concentrations with respect to other nutrients (i.e., nitrogen and phosphorus) may facilitate preferential growth of potentially harmful, non-silicious algae, with effects on water quality and ecosystem health (Conley et al. 1993; Turner et al. 2003). A recent synthesis of long-term trends in fluvial Si concentrations and loads from 60 streams showed that Si concentrations were changing, and that the timing of change during the year varied across and within biomes (Jankowski et al. 2023), indicating widespread changes in Si watershed processing may be occuring.

The seasonal pattern of stream water chemistry (i.e., its annual regime) reflects the integrated signal of climate, static watershed characteristics like lithology and terrestrial vegetation, and dynamic ecosystem processes such as instream and terrestrial primary productivity (Bernhardt et al. 2022; Bolotin et al. 2022). Recent efforts have characterized the fluvial regimes of major nutrient concentrations by evaluating the timing, magnitude, and rate of change in concentration peaks and troughs, which have provided powerful insights into controls on stream nutrient chemistry (Mattsson et al. 2015; Van Meter et al. 2020; Heiner et al. 2022). Previous work to understand stream Si seasonality has focused on temperate forest biomes, which are characterized by springtime concentration drawdowns driven by biological uptake with recovery throughout the summer and fall as uptake rates decline (Fulweiler and Nixon 2005; Carey and Fulweiler 2013) (Fig. 1a). Despite the importance of Si for freshwater and marine productivity, no published work has characterized Si regimes across biomes.

Existing work examining watershed cycling of Si has focused heavily on understanding hydrologic controls on Si concentrations. Concentration–discharge (CQ) relationships have shown that Si exhibits diluting (Carey et al. 2020), chemostatic (Godsey et al. 2009), and hysteretic (Hornberger et al. 2001) behavior, which indicates that fluvial Si concentrations are controlled by variable processes across landscapes. Little work has been done to assess drivers of different Si CQ patterns or to tie CQ behavior to seasonal variability in Si concentration.

A more complete understanding of Si regimes and their underlying drivers will provide insight into variability of Si regimes and help identify when and where Si concentrations may be changing most rapidly (Fig. 1b). Here, we take a step towards understanding the complex interactions that drive seasonal patterns of fluvial Si concentrations (i.e., regimes) by asking: 1) Are there distinct Si regimes among streams?; 2) Do streams show the same regime over their period of record?; and 3) How do climate, discharge, and source transport behavior influence regime pattern and stability (i.e., the tendency of a river to exhibit the same regime over time)? We hypothesize that rivers will exhibit distinct regimes (e.g., Bernhardt et al. 2022), that these regimes will cluster by climate zone (Li et al. 2022) and river size (Savoy et al. 2019), and that average CQ behavior (e.g., dilution and mobilization) will explain the timing of peaks and troughs (Minaudo et al. 2019). We further hypothesize that regimes will vary over time in response to interannual variability in discharge (Perdrial et al. 2014; Fazekas et al. 2021). To test these hypotheses and answer these questions, we analyzed data from 201 streams located across the Northern Hemisphere (Fig. 2a).

#### Methods

#### Data

We used dissolved Si concentration data sourced from published and/or publicly available datasets (Table S1; Johnson et al. 2023). Sites spanned eight climate zones between  $18^{\circ}$ N and  $70^{\circ}$ N. They varied in drainage area from < 1 km<sup>2</sup> to nearly 3 million km<sup>2</sup> and in mean stream discharge from < 0.01 cm to nearly 20,000 cm indicating a wide range in river size as



**Fig. 1.** (a) Conceptual model of factors influencing Si regime in temperate forests and (b) drivers that may lead to an unstable regime (movement in the blue arrow direction would lead to changes in timing of peaks and troughs, and movement in the red arrow direction would lead to dampening or exaggeration of regime variability).



**Fig. 2.** (a) Map of sites, colored by cluster membership. (b) Five clusters formed from normalized average site regimes colored by climate, and (c) mosaic plot showing the distribution of climates across the five clusters. The width of each column of the mosaic plot indicates the relative number of sites in each cluster, and the colored bars within each column represent the proportion of each cluster that is made up of each climate.

discharge and drainage area are positively correlated with stream width (Frasson et al. 2019). We limited our analyses to perennial streams with at least 5 yr of Si measurements and continuous streamflow (Text S1).

#### **Clustering of Si concentration**

We used the weighted regressions on time, discharge, and season (WRTDS) model (Hirsch et al. 2010) to estimate flow normalized (FN) Si concentrations at each site (Text S1). Daily FN concentrations were aggregated to create monthly FN Si concentrations. WRTDS model performance was evaluated by examining the flux bias statistic and the relationship between measured and modeled Si concentrations on all sampled days (Hirsch and De Cicco 2015) (Fig. S3).

To establish distinct Si regimes, we z-score normalized average monthly FN Si concentrations for each stream (i.e., one time series per stream) and performed a clustering analysis on these time series. We used dynamic time warping (DTW) to quantify the distance between average Si concentration curves for all streams in the dataset (i.e., each stream's time series was compared to every other stream's time series). DTW allows time series that do not align 1:1 in time but have similar shapes to be clustered together (refer to Sakoe and Chiba 1978 for details). We allowed time series to vary by +/-1 month around each time step to calculate their DTW distance (i.e., warping windowing of 1 month; Ratanamahatana and Keogh 2005) and clustered the sites using a partitional clustering algorithm with a prototyping function that iteratively refines cluster membership until convergence criteria are met (i.e., DTW barycenter averaging; Petitjean et al. 2011). We tested 2-15 total clusters (Text S2).

#### Stability analysis

To evaluate variability in regime membership within a given site, we generated regimes for each stream for every year over its period of record. These stream-year time series were then assigned to a cluster group based on the smallest DTW distance between the time series and the centroid of the clusters generated above (Text S2).

We developed a stability metric (*S*) to evaluate variability in regime membership within a given site. *S* can be broken down into two metrics: 1) how often a site deviated from its modal cluster over its period of record ( $S_T$ ; time stability) and 2) how many different clusters a site belonged to over its period of record ( $S_C$ ; cluster stability). Time and cluster stability provide different insights into the controls on Si regime at a given site. For example, we would expect different processes to control regime stability in a site that switches between two clusters over its period of record compared to a site that switches between all five clusters, even if they have similar values of time stability.  $S_T$  can be calculated as:

$$S_T = \frac{\gamma - \gamma_{\rm in}}{\gamma} \tag{1}$$

where *y* is the number of years in the period of record and  $y_{in}$  is the number of years where the annual regime is the same as the site's modal regime. We determined the modal cluster for each site based on its stream-year cluster membership.  $S_T$  varies between 0 (most stable) and 1 (least stable).  $S_C$  can be calculated as:

$$S_C = 1 + \frac{C - C_S}{C} \tag{2}$$

where *C* is the number of clusters and  $C_S$  is the number of clusters a given site belongs to over its period of record.  $S_C$  varies between 1 (least stable) and 1.8 (most stable). These metrics are combined to calculate overall stability (*S*):

$$S = 1 - \frac{S_T}{S_C} \tag{3}$$

Overall stability ranges from 0 (least stable) to 1 (most stable). Differences in stability between climate zones and clusters were assessed using a one-way analysis of variance (ANOVA) test followed by the Tukey honestly significant difference (HSD) test at a 95% confidence level.

#### Drivers

We evaluated whether climate, Si concentration, discharge, and concentration–discharge (*CQ*) behavior explained cluster membership and stability. We classified sites into Koeppen-Geiger subgroup climate zones (hereafter referred to as climates) using the *kgc* R package (Bryant et al. 2017) (Table S2). Si concentrations and discharge were averaged across all available data into a median value for each stream.

We evaluated the CQ relationship at each site using log  $C - \log Q$  slope (Eq. 4; Wymore et al. 2023). We calculated CQ slope from the raw concentration and discharge data. CQ relationships were analyzed using:

$$C = aQ^b \tag{4}$$

where *a* is a coefficient with units of concentration and *b* is an exponent that represents the slope of the log *C* – log *Q* relationship. Three dominant relationships can be inferred by the log *C* – log *Q* slope (Godsey et al. 2009): 1) chemostatic–insensitive relationship between *C* and *Q* (i.e., b = 0); 2) mobilization–positive relationship between *C* and *Q* (i.e., b > 0.1); and 3) dilution–negative relationship between *C* and *Q* (i.e., b < 0.1). *CQ* relationships can also be characterized as the ratio of the coefficients of variation (CV) of concentration and discharge (Eq. 5) (Thompson et al. 2011; Sullivan et al. 2019), where CV is defined as the standard deviation ( $\sigma$ ) normalized by its mean ( $\mu$ ):

$$\frac{\mathrm{C}\mathrm{V}_{c}}{\mathrm{C}\mathrm{V}_{Q}} = \frac{\mu_{Q}\sigma_{C}}{\mu_{C}\sigma_{Q}} \tag{5}$$

CVs can be interpreted as chemostatic or chemodynamic behavior when streams exhibit low  $(CV_c/CV_q \ll 1)$  or large concentration variability  $(CV_c/CV_q \ge 1)$ , respectively.

#### Results

#### Si regime clusters

We found five distinct cluster shapes that best explained the variability in Si concentration patterns (Fig. 2b) (Table S3). The cluster shapes were characterized by the timing of their minimum and maximum in the Northern Hemisphere: 1) fall peak, 2) fall trough, 3) spring trough, 4) spring trough-fall peak, and 5) spring trough-variable summer. The fall trough cluster contained the largest number of sites (68), whereas all other clusters contained between 30 and 40 sites. All clusters contained sites from at least four climates, and five of eight climates were found in at least three different clusters (Table S4).

#### Stability of cluster membership

There was a wide range (0.39–1.0) in overall cluster stability (*S*) across streams, and few significant differences in *S* between

climate and clusters. The humid tropical climate was least stable (mean S = 0.66), whereas the Mediterranean, humid temperate, and subarctic climates were most stable (S > 0.9) (Fig. 3c,d). The fall trough cluster was most stable (mean S = 0.93), and the spring trough-variable summer cluster was least stable (S = 0.80). In general, clusters with spring drawdown behavior had lower S than those without spring drawdowns.

### Association of cluster membership and stability with climate, discharge, and source-transport behavior

To identify drivers of cluster membership and stability, we assessed their relationships with climate, discharge, Si concentration, and CQ behavior. We found few significant differences in median Si concentration median Q among clusters (Fig. S1, Fig. S2a,b). We found no significant differentiation of CQ slope among clusters, and few significant differences in CVc/CVq values were found (Fig. S2c,d). All clusters exhibited a range of dilution, chemostatic, and mobilization behavior, with the strongest dilution observed in spring trough and fall trough clusters, and the strongest mobilization observed in fall trough and spring trough-variable summer clusters (Fig. 4). Although little direct relationship was found between climate and cluster, we did observe that climate emerged as a possible control of CQ behavior within a given cluster (Fig. 4).



**Fig. 3.** Overall stability (S) by cluster (**a**) and climate (**c**). Stability ranges between 0 and 1, where a value of 1 means that, for a given site, the seasonality curve was in the same cluster over its period of record. Climates are plotted in order of high to low mean annual temperature. Lower and upper lines of boxplot box are quartile 1 and 3, respectively. The middle line is the median. Vertical lines indicate minimum and maximum, if < +/-1.5\*interquartile range (IQR). Points outside +/-1.5\*IQR are considered outliers and are plotted above/below vertical lines. Points show all sites associated with each cluster (**a**) or climate (c). Matrix of significant differences between clusters (**b**) and climates (**d**) (Tukey HSD, alpha = 0.05). Abbreviations in b and d are defined in legends of a and c.



**Fig. 4.** CQ behavior (CQ slope on y-axis, CVc/CVq on x-axis) by cluster. Point color indicates climate, and size of point indicates coefficient of determination ( $R^2$ ) of the CQ slope.

We found dilution behavior in the fall trough and spring trough-variable summer clusters were dominated by subarctic streams whereas dilution behavior in the fall peak cluster was dominated by Mediterranean streams. In the same clusters, mobilization behavior dominated humid continental and humid subtropical streams.

#### Discussion

### Emergent regimes demonstrate variable Si seasonality across streams and over time

We found a great deal of variability in Si concentration regime across streams, with five main clusters capturing the range of behavior. Our conceptual model of the Si regime has been heavily informed by streams draining temperate forest basins, as those are systems where data availability is highest (Wall et al. 1998; Fulweiler and Nixon 2005; Carey and Fulweiler 2013). In these systems, spring declines in Si concentrations are driven primarily by diatom and terrestrial vegetation uptake, whereas fall peaks in Si concentrations occur due to lower biological activity and dissolution of fixed biogenic Si, a pattern that describes the spring trough cluster (Fig. 1a). Yet, our analysis of Si regimes across different biomes, including temperate forests, indicated that the humid continental climate (into which temperate forests fall) occurs in all five clusters indicating there is more regime variability even within a given climate than previous literature suggests. In addition, nearly 60% of streams showed some variability (stability < 1) in their regime, with over 20% of sites falling outside their modal behavior more than 25% of the time. Previous work describing biogeochemical regimes in rivers has focused primarily on describing "average conditions" (e.g., Bolotin et al. 2022); our work indicates that rivers exhibit multiple biogeochemical regimes. Our analyses indicate there is more variation in Si seasonality and stability than previously understood and point to a need to better understand intraannual and interannual Si dynamics across environmental gradients.

## Climate and hydrochemical drivers of cluster membership and stability

We found that cluster membership was not clearly explained by a single factor. Clusters were comprised from sites representing at least four different climates (Table S4), and distributions of median discharge and Si concentration were similar among clusters (Fig. S1). All five clusters showed dilution and chemostatic CQ behavior similar to the variability reported in the literature (e.g., Godsey et al. 2009; Carey et al. 2020), but we also observed mobilization behavior, which has been less frequently reported. To date Si mobilization behavior has been linked to agriculture during high flows that increase connectivity between the landscape and stream (Sethna et al. 2022), which may explain the



mobilization behavior observed at some of the analyzed sites. In contrast, we infer that snowmelt and distinct wet–dry seasonal cycles may drive the dilution behavior observed in subarctic and Mediterranean streams (Fig. 4). This indicates that although Si CQ may be related to climate and land use, it spans clusters indicating many CQ behaviors can give rise to similar regimes.

We hypothesize that differences among regimes likely arise from complex interactions between hydrologic and biologic processes and intraannual variability in solute transport behavior (Fig. 5). In diluting systems, spring troughs may be controlled by snowmelt (Fig. 5c-e; Carey et al. 2020), whereas fall peaks may be controlled by Si-rich baseflow (Fig. 5a,d; Benettin et al. 2015). However, in streams where winter flows are groundwater dominated. Si-rich baseflow may lead to a winter peak (Fig. 5b,c,e; Brooks et al. 2021). In contrast, in mobilizing streams peaks and troughs may arise due to variability in Si transport during the wet and dry season (Fig. 5ac,e; Wollheim et al. 2018). Seasonal patterns in streams that exhibit chemostatic behavior may be more controlled by biologic processes, which, like hydrologic processes, can exhibit variable timing. Vegetative uptake of Si occurs during the growing season but can vary in timing and magnitude among species (Fig. 5c,d; Epstein 1994), and algal blooms can occur throughout the spring and summer (Fig. 5b-e; Wall et al. 1998). These controls on Si concentrations can co-occur, leading to exaggeration or dampening of individual signals.

When examining drivers of stability, we found that regimes characterized by a spring trough generally had lower stability. Many studies have noted that climate change leads to earlier onset of the growing season (e.g., Sommer and Lengfellner 2008; Monahan et al. 2016) and increased variability in the timing of spring onset (Menzel et al. 2006). Changes in the timing or rate of Si drawdown, or variability in summer and fall seasonality, may lead to movement between spring drawdown regimes and therefore lower stability. For example, a site with consistent snow melt pulse but variable summer Si uptake may fluctuate between spring trough and spring trough-variable summer. Understanding stability is important because predicting the timing of Si delivery to downstream communities may be more difficult in sites with unstable behavior. Our work indicates that the fall peak

**Fig. 5.** Conceptual model of potential processes driving the patterns of the five Si regimes we observed. Average behaviors of all site seasonality curves are shown in black. Gray shading around the black lines represent error around the average regime shape (99% confidence interval). Non-italicized, blue text represents hydrologic processes that control Si concentrations, where dark blue represents dilution driven processes and royal blue represents mobilization driven processes. Italicized, orange text represents biologic processes that control Si concentrations, where light orange represent terrestrial processes and dark orange are aquatic processes. All sites are reflective of Northern Hemisphere seasonality.

and fall trough clusters and Mediterranean, subarctic, and humid temperate climates will be most consistent in Si export timing, whereas spring trough clusters and humid tropical or semiarid climates may exhibit more variable export timing.

#### **Implications and limitations**

Stability may be interpreted as an indicator of ecosystem resilience (Holling 1973), where sites with more stable regimes may be considered more resilient (i.e., able to withstand perturbations). Our work indicates that spring trough sites are less stable, possibly reflecting the impact of climate change on spring snowmelt and Si uptake processes. We observed that some sites show systematic shifts from one regime to another (e.g., directional), whereas other sites varied between all five regimes (e.g., chaotic). We hypothesize that these two behaviors should be interpreted differently. Sites that exhibit more directional instability (i.e., shift from one regime to another) may be interpreted as being driven by mechanistic processes (e.g., reduction in groundwater input, earlier snowmelt) (Spence et al. 2015; Foks et al. 2018; Zhi et al. 2020), whereas sites with chaotic instability (i.e., shift randomly among many regimes) may suggest that systems have weaker internal feedback mechanisms and thus shift rapidly in response to changes in external drivers (Dent et al. 2002), such as variable onset of Si uptake (Menzel et al. 2006), weather patterns (i.e., floods and droughts; Greenwood and Eimers 2023), or variability in stream-release schedules (Sawyer et al. 2009). Future work should be done to evaluate where directional and chaotic instability occur and evaluate mechanistic drivers of directional shifts.

Four additional considerations should be made to better understand Si regime behavior and its relationship to ecosystem resilience. First, regimes should be evaluated for ecological relevance. About 5% of our sites had an annual CVc < 0.1, indicating that variability in Si concentration was low (Fig. S5). At these sites, regimes may reflect amplified differences between the minimum and maximum Si concentrations due to z-score normalization, and therefore may not be ecologically significant. Second, Si regimes need to be related to other controlling factors including basin characteristics (e.g., basin slope and reservoir dynamics), climate (e.g., precipitation and temperature), and light availability. Third, processes that explain the behavior of Si regimes may be influenced by other nutrient cycles; thus, examining many nutrient regimes in tandem may be key to predicting aquatic productivity regimes and how they shift over time. Finally, the data used in this work represent streams primarily from North America and Europe, which are estimated to deliver only 16% of total global Si export to oceans (Dürr et al. 2011). Expanding our study of underrepresented regions and climates may elucidate other Si regimes not captured in this analysis.

#### References

- Benettin, P., S. W. Bailey, J. L. Campbell, M. B. Green, A. Rinaldo, G. E. Likens, K. J. McGuire, and G. Botter. 2015. Linking water age and solute dynamics in streamflow at the Hubbard brook experimental Forest, NH, USA. Water Resour. Res. **51**: 9256–9272. doi:10.1002/2015WR017552
- Bernhardt, E. S., and others. 2022. Light and flow regimes regulate the metabolism of rivers. Proc. Natl. Acad. Sci. USA 119: e2121976119. doi:10.1073/pnas.2121976119
- Bolotin, L. A., B. M. Summers, P. Savoy, and J. R. Blaszczak. 2022. Classifying freshwater salinity regimes in central and western U.S. streams and rivers. Limnol. Oceanogr. Lett. 8: 103–111. doi:10.1002/lol2.10251
- Brooks, P. D., and others. 2021. Groundwater-mediated memory of past climate controls water yield in snowmeltdominated catchments. Water Resour. Res. 57: e2021WR030605. doi:10.1029/2021WR030605
- Bryant, C., N. Wheeler, F. Rubel, and R. French. 2017. *kgc: Koeppen-Geiger climatic zones*. R package version 1.0.0.2, https://CRAN.R-project.org/package=kgc
- Buesseler, K. O. 1998. The decoupling of production and particulate export in the surface ocean. Global Biogeochem. Cycles 12: 297–310. doi:10.1029/97GB03366
- Carey, J. C., and R. W. Fulweiler. 2012. Human activities directly alter watershed dissolved silica fluxes. Biogeochemistry **111**: 125–138. doi:10.1007/s10533-011-9671-2
- Carey, J. C., and R. W. Fulweiler. 2013. Watershed land use alters riverine silica cycling. Biogeochemistry **113**: 525–544. doi:10.1007/s10533-012-9784-2
- Carey, J. C., J. Gewirtzman, S. E. Johnston, A. Kurtz, J. Tang, A. M. Vieillard, and R. G. M. Spencer. 2020. Arctic River dissolved and biogenic silicon exports—Current conditions and future changes with warming. Global Biogeochem. Cycles 34: e2019GB006308. doi:10.1029/2019GB006308
- Conley, D. J., C. L. Schelske, and E. F. Stoermer. 1993. Modification of the biogeochemical cycle of silica with eutrophication. Mar. Ecol. Prog. Ser. 101: 179–192.
- Deng, K., S. Yang, and Y. Guo. 2022. A global temperature control of silicate weathering intensity. Nat. Commun. 13: 1–10. doi:10.1038/s41467-022-29415-0
- Dent, C. L., G. S. Cumming, and S. R. Carpenter. 2002. Multiple states in river and lake ecosystems. Philos. Trans. R. Soc. Lond. B Biol. Sci. **357**: 635–645. doi:10.1098/rstb. 2001.0991
- Dürr, H. H., M. Meybeck, J. Hartmann, G. G. Laruelle, and V. Roubeix. 2011. Global spatial distribution of natural riverine silica inputs to the coastal zone. Biogeosciences 8: 597– 620. doi:10.5194/bg-8-597-2011
- Epstein, E. 1994. The anomaly of silicon in plant biology. Proc. Natl. Acad. Sci. USA **91**: 11–17. doi:10.1073/pnas.91. 1.11
- Fazekas, H. M., W. H. McDowell, J. B. Shanley, and A. S. Wymore. 2021. Climate variability drives watersheds

along a transporter-transformer continuum. Geophys. Res. Lett. **48**: e2021GL094050. doi:10.1029/2021GL09 4050

- Foks, S. S., E. G. Stets, K. Singha, and D. W. Clow. 2018. Influence of climate on alpine stream chemistry and water sources. Hydrol. Process. **32**: 1993–2008. doi:10.1002/hyp. 13124
- Frasson, R. P. D. M., T. M. Pavelsky, M. A. Fonstad, M. T. Durand, G. H. Allen, G. Schumann, C. Lion, R. E. Beighley, and X. Yang. 2019. Global relationships between river width, slope, catchment area, meander wavelength, sinuosity, and discharge. Geophys. Res. Lett. 46: 3252–3262. doi: 10.1029/2019GL082027
- Fulweiler, R. W., and S. W. Nixon. 2005. Terrestrial vegetation and the seasonal cycleof dissolved silica in a southern new Englandcoastal river. Biogeochemistry 74: 115–130. doi:10. 1007/s10533-004-2947-z
- Godsey, S. E., J. W. Kirchner, and D. W. Clow. 2009. Concentration–discharge relationships reflect chemostatic characteristics of US catchments. Hydrol. Process. 23: 1844–1864. doi:10.1002/hyp.7315
- Greenwood, W. J., and M. C. Eimers. 2023. Hydroclimatic variability across the international Lake of the Woods watershed: Implications for nutrient export and climate sensitivity. J. Great Lakes Res. **49**: 8–20. doi:10.1016/j.jglr. 2022.12.004
- Gregg, W. W., and C. S. Rousseaux. 2014. Decadal trends in global pelagic ocean chlorophyll: A new assessment integrating multiple satellites, in situ data, and models. J. Geophys. Res. Oceans 119: 5921–5933. doi:10.1002/201 4JC010158
- Heiner, M., M. J. Heaton, B. Abbott, P. White, C. Minaudo, and R. Dupas. 2022. Model-based clustering of trends and cycles of nitrate concentrations in Rivers across France. J. Agric. Biol. Environ. Stat. 28: 74–98. doi:10.1007/ s13253-022-00513-2
- Hirsch, R. M., D. Moyer, and S. A. Archfield. 2010. Weighted regressions on time, discharge, and season (WRTDS), with an application to Chesapeake Bay river inputs. J. Am. Water Resour. Assoc. **46**: 857–880. doi:10.1111/j.17 52-1688.2010.00482.x
- Hirsch, R. M., & De Cicco, L. A. (2015). User guide to Exploration and Graphics for RivEr Trends (EGRET) and dataRetrieval: R packages for hydrologic data (USGS Numbered Series 4-A10); Chapter-A10, p. 104. U.S. Geological Survey 10.3133/tm4A10
- Holling, C. S. 1973. Resilience and stability of ecological systems. Annu. Rev. Ecol. Syst. 4: 1–23. doi:10.1146/annurev. es.04.110173.000245
- Hornberger, G. M., T. M. Scanlon, and J. P. Raffensperger. 2001. Modelling transport of dissolved silica in a forested headwater catchment: The effect of hydrological and chemical time scales on hysteresis in the concentration– discharge relationship. Hydrol. Process. **15**: 2029–2038. doi:10.1002/hyp.254

- Jankowski, K. J., and others. 2023. Long-term change in the concentration and yield of riverine dissolved silicon from the poles to the tropics. Global Biogeochem. Cycles **37**: e2022GB007678. doi:10.1029/2022GB007678
- Johnson, K., and others. 2023. *Monthly dissolved silicon concentrations from 198 rivers in the northern hemisphere*. US Geological Survey. doi:10.5066/P9CXDEE8
- Li, L., B. Stewart, W. Zhi, K. Sadayappan, S. Ramesh, D. Kerins, G. Sterle, A. Harpold, and J. Perdrial. 2022. Climate controls on river chemistry. Earths Future 10: e2021EF002603. doi:10.1029/2021EF002603
- Maguire, T. J., and R. W. Fulweiler. 2016. Urban dissolved silica: Quantifying the role of groundwater and runoff in wastewater influent. Environ. Sci. Technol. **50**: 54–61. doi: 10.1021/acs.est.5b03516
- Malviya, S., and others. 2016. Insights into global distribution and diversity in the worlds ocean. Proc. Natl. Acad. Sci. USA **113**: E1516–E1525.
- Mattsson, T., P. Kortelainen, A. Räike, A. Lepistö, and D. N. Thomas. 2015. Spatial and temporal variability of organic C and N concentrations and export from 30 boreal rivers induced by land use and climate. Sci. Total Environ. **508**: 145–154. doi:10.1016/j.scitotenv.2014.11.091
- Menzel, A., T. H. Sparks, N. Estrella, and D. B. Roy. 2006. Altered geographic and temporal variability in phenology in response to climate change. Glob. Ecol. Biogeogr. **15**: 498–504. doi:10.1111/j.1466-822X.2006.00247.x
- Minaudo, C., R. Dupas, C. Gascuel-Odoux, V. Roubeix, P.-A. Danis, and F. Moatar. 2019. Seasonal and event-based concentration-discharge relationships to identify catchment controls on nutrient export regimes. Adv. Water Resour. 131: 103379. doi:10.1016/j.advwatres.2019.103379
- Monahan, W. B., A. Rosemartin, K. L. Gerst, N. A. Fisichelli, T. Ault, M. D. Schwartz, J. E. Gross, and J. F. Weltzin. 2016. Climate change is advancing spring onset across the US national park system. Ecosphere 7: e01465. doi:10.1002/ecs2.1465
- Perdrial, J. N., and others. 2014. Stream water carbon controls in seasonally snow-covered mountain catchments: Impact of inter-annual variability of water fluxes, catchment aspect and seasonal processes. Biogeochemistry **118**: 273–290. doi:10.1007/s10533-013-9929-y
- Petitjean, F., A. Ketterlin, and P. Gançarski. 2011. A global averaging method for dynamic time warping, with applications to clustering. Pattern Recognition 44: 678–693. doi: 10.1016/j.patcog.2010.09.013
- Ratanamahatana, C. A., and E. Keogh. 2005. Three myths about dynamic time warping data mining, p. 506–510. *Proceedings of the 2005 SIAM International Conference on Data Mining (SDM)*. Society for Industrial and Applied Mathematics. doi:10.1137/1.9781611972757.50
- Sakoe, H., and S. Chiba. 1978. Dynamic programming algorithm optimization for spoken word recognition. IEEE

Wiley Online

Library for rules of use; OA articles

are governed by the applicable Creative Commons

#### Johnson et al.

Establishing fluvial Si regimes

Trans. Acoust. Speech Signal Process. **26**: 43–49. doi:10. 1109/TASSP.1978.1163055

- Savoy, P., A. P. Appling, J. B. Heffernan, E. G. Stets, J. S. Read, J. W. Harvey, and E. S. Bernhardt. 2019. Metabolic rhythms in flowing waters: An approach for classifying river productivity regimes. Limnol. Oceanogr. 64: 1835–1851. doi:10. 1002/lno.11154
- Sawyer, A. H., M. B. Cardenas, A. Bomar, and M. Mackey. 2009. Impact of dam operations on hyporheic exchange in the riparian zone of a regulated river. Hydrol. Process. 23: 2129–2137. doi:10.1002/hyp.7324
- Sethna, L. R., T. V. Royer, S. L. Speir, M. T. Trentman, U. H. Mahl, L. P. Hagemeier, and J. L. Tank. 2022. Silicon concentrations and stoichiometry in two agricultural watersheds: Implications for management and downstream water quality. Biogeochemistry 159: 265–282. doi:10.1007/ s10533-022-00927-7
- Sommer, U., and K. Lengfellner. 2008. Climate change and the timing, magnitude, and composition of the phytoplankton spring bloom. Glob. Chang. Biol. **14**: 1199–1208. doi:10.1111/j.1365-2486.2008.01571.x
- Spence, C., S. V. Kokelj, S. A. Kokelj, M. McCluskie, and N. Hedstrom. 2015. Evidence of a change in water chemistry in Canada's subarctic associated with enhanced winter streamflow. J. Geophys. Res. Biogeo. **120**: 113–127. doi:10. 1002/2014JG002809
- Struyf, E., and others. 2010. Historical land use change has lowered terrestrial silica mobilization. Nat. Commun. 1: 129. doi:10.1038/ncomms1128
- Sullivan, P. L., M. W. Stops, G. L. Macpherson, L. Li, D. R. Hirmas, and W. K. Dodds. 2019. How landscape heterogeneity governs stream water concentration-discharge behavior in carbonate terrains (Konza prairie, USA). Chem. Geol. 527: 118989. doi:10.1016/j.chemgeo.2018.12.002
- Thompson, S. E., N. B. Basu, J. Lascurain Jr., A. Aubeneau, and P. S. C. Rao. 2011. Relative dominance of hydrologic versus biogeochemical factors on solute export across impact gradients. Water Resour. Res. 47: 1–20. doi:10.1029/2010WR 009605
- Turner, R. E., Rabalais, N. N., Justic', D., & Dortch, Q. (2003). Global patterns of dissolved N, P and Si in large rivers. Biogeochemistry, 64(3), 297–317. 10.1023/A:1024960007569
- Van Meter, K. J., S. Chowdhury, D. K. Byrnes, and N. B. Basu. 2020. Biogeochemical asynchrony: Ecosystem drivers of seasonal concentration regimes across the Great Lakes

Basin. Limnol. Oceanogr. 65: 848-862. doi:10.1002/lno. 11353

- Wall, G. R., P. J. Phillips, and K. Riva-Murray. 1998. Seasonal and spatial patterns of nitrate and silica concentrations in Canajoharie Creek, New York. J. Environ. Qual. 27: 381– 389. doi:10.2134/jeq1998.00472425002700020019x
- Wollheim, W. M., and others. 2018. River network saturation concept: Factors influencing the balance of biogeochemical supply and demand of river networks. Biogeochemistry 141: 503–521. doi:10.1007/s10533-018-0488-0
- Wymore, A. S., and others. 2023. Revisiting the origins of the power-law analysis for the assessment of concentrationdischarge relationships. Water Resour. Res. **59**: e2023WR034910. doi:10.1029/2023WR034910
- Zhi, W., Williams, K. H., Carroll, R. W. H., Brown, W., Dong, W., Kerins, D., & Li, L. (2020). Significant stream chemistry response to temperature variations in a highelevation mountain watershed. Commun. Earth Environ., 1(1), 43.

#### Acknowledgments

This work was completed as part of a synthesis working group entitled: From Poles to Tropics: A multibiome synthesis investigating the controls on river Si exports and was supported through the Long Term Ecological Research Network Office (LNO) (NSF award numbers 1545288 and 1929393) and the National Center for Ecological Analysis and Synthesis, UCSB awarded to KJJ and JCC. KJJ was supported by the US Army Corps of Engineers Upper Mississippi River Restoration Program. Funding was provided to J. Carey from the Babson Faculty Research Fund and the Andrew J. Butler and Debi Butler Term Chair. Funding was provided to K. Johnson and P.L. Sullivan by the Department of Energy (Grant No. DE-SC0020146) and the National Science Foundation (NSF: 2034232 and 2012796). ASW received support from the National Science Foundation and EPSCoR project Canary in the Watershed (NSF EPS-1929148). Support was also provided by the New Hampshire Agricultural Experiment Station. This work was supported by the USDA National Institute of Food and Agriculture Hatch Multi-State Project 1022291 (ASW), 0225006 (WMW) and McIntire-Stennis Project 1019522 (WHM). This is scientific contribution 2971. A. Poste was supported by the 'Catchment to Coast' research programme at the Fram Centre for High North Research (PL Poste). Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government.

> Submitted 03 May 2023 Revised 15 November 2023 Accepted 27 November 2023